**f** 



# Water Resources Research<sup>®</sup>

## **RESEARCH ARTICLE**

10.1029/2021WR031871

#### **Key Points:**

- Uncertainties between pedotransfer functions are comparable to uncertainties in soil texture
- Pedotransfer function choice has large effect on hydrological and smaller effects on ecosystem dynamics
- Complex topography amplifies the importance of pedotransfer function uncertainties

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

A. Paschalis, a.paschalis@imperial.ac.uk

#### **Citation:**

Paschalis, A., Bonetti, S., Guo, Y., & Fatichi, S. (2022). On the uncertainty induced by pedotransfer functions in terrestrial biosphere modeling. *Water Resources Research*, 58, e2021WR031871. https://doi. org/10.1029/2021WR031871

Received 21 DEC 2021 Accepted 15 AUG 2022

#### **Author Contributions:**

Conceptualization: Athanasios Paschalis, Sara Bonetti, Simone Fatichi Data curation: Athanasios Paschalis Formal analysis: Athanasios Paschalis, Yanran Guo

Investigation: Athanasios Paschalis Methodology: Athanasios Paschalis Software: Athanasios Paschalis, Simone Fatichi

Validation: Athanasios Paschalis Visualization: Athanasios Paschalis Writing – original draft: Athanasios Paschalis, Sara Bonetti, Simone Fatichi Writing – review & editing: Athanasios Paschalis, Sara Bonetti, Simone Fatichi

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## On the Uncertainty Induced by Pedotransfer Functions in Terrestrial Biosphere Modeling

#### Athanasios Paschalis<sup>1</sup>, Sara Bonetti<sup>2</sup>, Yanran Guo<sup>1</sup>, and Simone Fatichi<sup>3</sup>

<sup>1</sup>Department of Civil and Environmental Engineering, Imperial College London, London, UK, <sup>2</sup>School of Architecture, Civil and Environmental Engineering, École Polytechnique Fédérale de Lausanne, Lausanne, Switzerland, <sup>3</sup>Department of Civil and Environmental Engineering, National University of Singapore, Singapore, Singapore

**Abstract** Hydrological, ecohydrological, and terrestrial biosphere models depend on pedotransfer functions for computing soil hydraulic parameters based on easily measurable variables, such as soil textural and physical properties. Several pedotransfer functions have been derived in the last few decades, providing divergent estimates of soil hydraulic parameters. In this study, we quantify how uncertainties embedded in using different pedotransfer functions propagate to ecosystem dynamics, including simulated hydrological fluxes and vegetation response to water availability. Using a state-of-the-art ecohydrological model applied at 79 sites worldwide, we show that uncertainties related to pedotransfer functions can affect both hydrological and vegetation dynamics. Uncertainties in evapotranspiration, plant productivity, and vegetation structure, quantified as leaf area, are in the order of  $\sim 10\%$  at annual time scales. Runoff and groundwater recharge uncertainties are one order of magnitude larger. All uncertainties are largely amplified when small-scale topography is taken into account in a distributed domain, especially for water-limited ecosystems with low permeability soils. Overall, pedotransfer function related uncertainties for a given soil type are higher than uncertainties across soil types in both hydrological and ecosystem dynamics. The magnitude of uncertainties is climate-dependent but not soil type-dependent. Evapotranspiration, vegetation structure, and plant productivity uncertainties are higher in water-limited semiarid climates, whereas groundwater recharge uncertainties are higher in climates where potential evapotranspiration is comparable to precipitation.

**Plain Language Summary** Models that simulate the water and carbon cycles at the land surface need to consider the properties of soil that affect water movement. As those properties are not easily observable, models typically rely on empirical functions for their estimation. These are called pedotransfer functions, which use easily measurable quantities, such as the description of soil texture, to predict less easily measurable soil hydraulic properties. In this study, we quantified how differences in many of those pedotransfer functions introduce uncertainties in the way ecosystem water and carbon cycles are simulated. We show that differences in soil hydraulic properties estimation can introduce significant uncertainty in both the water and carbon cycles. The uncertainties in the water cycle are largest for the prediction of water flowing toward rivers and groundwater replenishment rates. The uncertainties in the carbon cycle are largest in semiarid climates. All uncertainties are further amplified by the presence of prominent topographic features (e.g., valleys and mountains) that introduce further complexity in relation to soil water movement.

#### 1. Introduction

Terrestrial ecosystem and ecohydrological models are of paramount importance for understanding and quantifying the coupled water and carbon cycles. Despite their physics-based modeling approach, several sources of uncertainties affect the robustness of their predictions (e.g., Fisher et al., 2014; Zhou et al., 2018). Uncertainties include epistemic sources due to specific model formulations (e.g., Schwalm et al., 2019) and poorly known model parameters (e.g., Pappas et al., 2013) due to the lack of high fidelity in situ data at commensurate scales. One of the major processes that affects both the water and carbon cycles is the movement of water in the soil. Uncertainties involved in modeling soil water transport may impact key hydrological fluxes, such as runoff generation, groundwater recharge, and evapotranspiration, and, in turn, changes in water content in the root zone may influence vegetation dynamics, especially in areas with limited water availability.

Most current generation ecohydrological and terrestrial ecosystem models simulate water movement in soils by solving either the 1-D or the 3-D Richards equation, adopting different numerical schemes (e.g., Farthing &

Ogden, 2017; Fatichi, Vivoni, et al., 2016). A major source of uncertainty when solving the Richards equation for the variably saturated critical zone (the region where vegetation typically accesses water) is the highly nonlinear relationship between soil water content and water potential (i.e., the soil water retention curve) and the dependence of hydraulic conductivity on the degree of saturation (i.e., the hydraulic conductivity function) (Assouline & Or, 2013). Models typically adopt the Brooks and Corey (BC) or the Mualem-Van Genuchten (MvG) formulations (Brooks & Corey, 1964; Mualem, 1976; Van Genuchten, 1980) to describe these nonlinear relationships. The estimation of model parameters for the BC and MvG models requires extensive data collection and is commonly performed, in terrestrial biosphere models, by means of pedotransfer functions (PTFs) (e.g., Assouline & Or, 2013; Patil & Singh, 2016; Van Looy et al., 2017; Vereecken et al., 2010). PTFs are essentially empirical models that use easy-to-measure soil properties (mostly soil texture, bulk density, and organic content) to predict the saturated hydraulic conductivity and the parameters of either the BC or the MvG formulations. PTFs use different statistical techniques, from multiple linear regression to advanced machine learning, to obtain those parameters based on a generally limited set of observations (e.g., Patil & Singh, 2016; Van Looy et al., 2017; Wösten et al., 2001; Y. Zhang & Schaap, 2019). Some soil textures (e.g., clays) are typically under-sampled and observations are usually obtained within a limited geographic region, leading to highly variable predictions across PTFs trained with different data sets (e.g., Wagner et al., 2001) and limited transferability across geographic locations (Gupta et al., 2020; Vereecken et al., 2016). The use of PTFs in both land surface and terrestrial biosphere modeling is ubiquitous (e.g., Cooper et al., 2020; Van Looy et al., 2017; Vereecken et al., 2016), and thus understanding the implications of their use is crucial. Global scale application of such models, in particular, rely heavily on PTFs (e.g., Dai et al., 2019), as specific soil hydraulic properties are available only locally.

The disagreement across PTF predictions is expected to impact hydrological dynamics, and this has been confirmed by a number of studies, where soil moisture (e.g., Deng et al., 2009; Loosvelt et al., 2011; Pinnington et al., 2021), evapotranspiration (e.g., Weihermüller et al., 2021), runoff production, and flood generation have been found to be strongly affected by soil hydraulic parameters (e.g., Mohajerani et al., 2021; W. Sun et al., 2016). The uncertainty introduced by PTFs in the hydrological dynamics has also been found to be modulated by small scale topography (e.g., Mohajerani et al., 2021).

While previous research has highlighted the importance of the variability introduced by PTFs in hydrological and land surface responses (e.g., Chirico et al., 2010; Mohajerani et al., 2021), there has been no comprehensive analysis on how those uncertainties could propagate from the water cycle to the carbon cycle via uncertainties in the predicted plant water stress, which limits plant water uptake and photosynthesis. Specifically, little attention has been given on how different shapes of the water retention curve, as well as differences in hydraulic conductivity, affect soil water flow and, subsequently, plant water availability and vegetation dynamics. Additionally, most land surface and ecosystem model applications, which are essential for climate change projections, neglect the role of topographic features as they consider one-dimensional vertical domains, thus potentially misrepresenting the importance of soil hydraulic properties (see also Fatichi et al. (2020)).

In this study, by using a state-of-the-art ecohydrological/terrestrial biosphere model with a dynamic vegetation component, we quantify how disagreements between PTF predictions lead to uncertainties in the dynamics of the water and carbon cycles at 79 locations worldwide, covering most terrestrial biomes and soil types. The analysis is first carried out for one-dimensional domains to mimic the common use of these models at global or large scales. Subsequently, we take small scale topography explicitly into account by simulating a distributed domain of 4 km<sup>2</sup> with a 40 × 40 m<sup>2</sup> resolution. The specific hypotheses of the study are:

- 1. Uncertainties related to the choice of PTFs are comparable to uncertainties in soil texture itself.
- 2. Uncertainties are overall larger for the water cycle than the carbon cycle.
- 3. Uncertainties in hydrological dynamics are larger in wet climates, and uncertainties in ecosystem vegetation dynamics are larger in areas where vegetation is limited by water availability (i.e., arid/semiarid regions).
- 4. Uncertainties for all ecosystem dynamics are larger for clay rich soils, where the training set of PTFs is commonly under-sampled.
- 5. Small scale topography magnifies the uncertainties generated by the PTF choice as it allows cumulative effects through subsurface lateral water flow.

#### 2. Materials and Methods

#### 2.1. Numerical Experiment

#### 2.1.1. Model Description

Numerical simulations are carried out with the ecohydrological and terrestrial biosphere model T&C (e.g., Fatichi et al., 2012, 2021; Fatichi, Manzoni, et al., 2019; Mastrotheodoros et al., 2020; Paschalis et al., 2015, 2018; Z. Zhang et al., 2021). T&C computes the water, carbon and energy balances at the land surface, and can be run as a distributed model on a regular grid geometry, taking into account small scale topography.

Specifically, in T&C, soil water transport is simulated using a quasi-3-D formulation of the Richards equation. Vertical soil water fluxes are simulated by solving the 1-D Richards equation using a finite volume approach. Horizontal soil water fluxes are approximated using a kinematic wave approximation (Fatichi et al., 2012; Hopp et al., 2016). Surface water flow, including both hillslope and channel routing, is simulated using the kinematic wave approximation with a time adaptive finite volume discretization and a D8 downhill flow direction.

Evapotranspiration is simulated using the common resistor analogue scheme (Bonan, 2019). Resistances that are considered in T&C include atmospheric, leaf boundary, undercanopy, and soil resistances. Soil resistance affects soil evaporation and it is simulated using the formulation of Shahraeeni et al. (2012). Leaf stomatal resistance that affects plant transpiration is simulated using the Leuning-based formulation (Paschalis et al., 2017; Tuzet et al., 2003).

Soil water dynamics affect the carbon cycle through their impact on limiting photosynthesis during periods of low plant water availability. Photosynthetic rates, modeled according to Collatz et al. (1991, 1992), are limited when the root zone integrated soil water potential drops below a plant-specific threshold. This threshold is a flexible parameter in T&C and can be tuned to represent any vegetation species. Photosynthesis suppression with decreasing water potentials follows a sigmoidal function (Fatichi et al., 2012). Prolonged duration of high vegetation stress can trigger leaf shedding in T&C, based on a plant specific stress duration threshold. When photosynthesis is limited, plant stomata close and stomatal conductance (i.e., the reciprocal of resistance) is reduced linearly with the reduction of leaf-scale photosynthesis. Vegetation structure affects hydrological fluxes via the dependence of interception, throughfall, and transpiration on leaf area.

The carbon assimilated through photosynthesis is either used for autotrophic respiration, which is temperature dependent, or used for plant growth by being allocated in the model's plant biomass pools (leaves, living sapwood, fine roots, carbohydrate reserves, fruits and flowers, and heartwood). Plant phenology is simulated using empirical formulae, where soil water stress, root zone temperature, solar radiation, and photoperiod are the factors leading to changes in the plant's phenological state.

In T&C both the BC and MvG soil hydraulic functions can be used. The modification of Webb (2000) is implemented for the MvG model to improve its predictions at low values of soil water content. In T&C the saturated hydraulic conductivity and the parameters describing the shape of the water retention curve can be derived by PTFs. In this study, seven different PTFs were used (Figure 1, Figure S2 in Supporting Information S1): three to estimate the parameters for the BC model (Cosby et al., 1984; K. Saxton et al., 1986; K. E. Saxton & Rawls, 2006) and four for the MvG model (Weynants et al., 2009; Wösten et al., 1999; Y. Zhang & Schaap, 2017; Zacharias & Wessolek, 2007). All PTFs, apart from that of Zacharias and Wessolek (2007), provide values for both the saturated hydraulic conductivity and the parameters for the shape of the water retention curve. For the case of Zacharias and Wessolek (2007) PTF, estimates for the saturated hydraulic conductivity were obtained by the PTF of K. E. Saxton and Rawls (2006). All PFTs depend on a subset of the following soil properties: percentage of clay, sand, silt, organic material, and bulk density.

Processes related to soil biogeochemistry and belowground carbon, nitrogen, phosphorus and potassium cycles are not simulated in this study, even though the model we used has this capability (Fatichi, Manzoni, et al., 2019).

#### 2.1.2. Plot Scale Simulations

The first set of simulations performed here neglects topography and uses T&C at the plot scale, for example, a one-dimensional vertical model over a flat domain, similarly to global scale applications of Earth system models. The model was set up for 79 sites globally, covering a wide range of climates and biomes (Table S1 in Supporting Information S1). For these sites, the model has been previously calibrated and validated (Fatichi, Leuzinger,



## Water Resources Research



Figure 1. (a) Boxplot of saturated hydraulic conductivity values estimated by six pedotransfer functions (PTFs) (all except the PTF) of Zacharias and Wessolek (2007), which does not provide estimates for hydraulic conductivity) across the 12 USDA soil texture classes (colored dots show values obtained from the different PTFs). For each PFT class one soil texture was used based on Table 1 and the range represents the effect of the PTFs only. Same as (a) but for the soil water matric potential at 50% effective saturation.

et al., 2016; Fatichi & Pappas, 2017; Moustakis et al., 2022; Paschalis et al., 2018). To run the model, we used hourly local weather data including air temperature, short and longwave radiation, humidity, cloudiness, and wind speed. The time span of data availability was between 2 and 32 years, with an average simulation length of 9.3 years.

For all 79 sites, two sets of simulations were carried out. For the first simulation set, which was used for the quantification of uncertainties due to PTF differences in ecosystem functioning, local soil properties (soil texture, organic content, and bulk density) were used, and the model was run using all seven PTFs. The second simu-

 Table 1

 Soil Texture, Organic Content, and Bulk Density for the 12 USDA Soil
 Classes

	% Sand	% Clay	% Silt	% Organic	Bulk density [gcm <sup>-3</sup> ]
Silt	10	5	83	2	1.44
Silty clay loam	10	30	58	2	1.34
Silty clay	10	45	43	2	1.27
Silt loam	20	15	63	2	1.42
Clay	20	60	18	2	1.26
Clay loam	30	35	33	2	1.40
Loam	40	20	38	2	1.46
Sandy clay	50	40	8	2	1.48
Sandy clay loam	60	30	8	2	1.52
Sandy loam	65	10	23	2	1.49
Loamy sand	80	5	13	2	1.46
Sand	90	5	3	2	1.45

lation set was used to compare the within soil-type uncertainties that are introduced by the choice of PTFs with the across soil-type uncertainties that are introduced by lack of knowledge of the exact soil physical properties. To this purpose, for each site  $12 \times 7$  simulations were performed where 12 different soil types, representative of each USDA class, were used (Table 1), together with all 7 PTFs, resulting in a total of 6,636 simulations. In all soil textures, we added a 2% fraction of soil organic matter, a texture not included in the USDA classification. In this sense, our definition for each class is not identical to USDA but very similar (classification details are presented in Table 1). For the PTFs that did not include an organic matter fraction in the soil texture, this 2% fraction was considered as silt.

#### 2.1.3. Distributed Simulations

For two of the 79 sites, Konza and Oak Ridge (ORNL), representative of a water-limited grassland and a temperate deciduous forest, respectively, the distributed version of T&C was used to quantify how spatially distributed water flow dynamics in a complex topography alters PTF-related uncertainties. Specifically, the model was run for both sites using an idealized 4 km 2 steep catchment topography (Figure 2) with a 290 m elevation difference draining to a single outlet. Land use was considered identical for the entire domain (i.e., no spatial heterogeneity). All other properties (e.g., plant





Figure 2. (a) Elevation (m), (b) slope (m/m), and (c) topographic index for the idealized catchment used in the distributed simulations.

species composition and their plant-related parameters) were identical to the plot scale simulations for both sites. Previous research has used the same synthetic catchment (Ivanov et al., 2008; Remondi et al., 2019). The spatial discretization of the model was 40 m, leading to 2,500 computational cells.

For both sites, two soil types were used. One representative of a fast draining sandy soil (sandy soil: 40% sand, 5% clay, 2% organic, and a bulk density of 1.47 gcm<sup>-3</sup>) and one of a slow draining clay soil (10% sand, 30% clay, 2% organic, and a bulk density of 1.34 gcm<sup>-3</sup>). The soil depth was identical to the soil depth used for the point scale simulations for those two sites (Konza: 2,500 mm, ORNL: 1,500 mm). For the two sites and the two soil types, a 4-year long simulation was performed using all 7 PTFs. The soil type and soil depth were considered homogeneous across the entire catchment and an anisotropy ratio of 50 between the horizontal and vertical hydraulic conductivities was considered for all simulations (e.g., Assouline & Or, 2006). An additional simulation with an anisotropy ratio of 10 (not shown here) was also performed to assess the sensitivity of our simulations to this parameter, which turned out to be of low importance and thus not further discussed. A zero water flux boundary condition at the soil bottom was used in the spatially distributed simulations to avoid additional uncertainties related to resolving bedrock groundwater dynamics, by neglecting flow at the soil-bedrock interface. Accompanying the spatially distributed simulations, point scale simulations with identical meteorological forcing and soil properties (including the zero flux boundary condition) were performed for those two sites to facilitate comparison and allow a proper quantification of the sensitivity of accounting for topography and distributed water fluxes.

#### 2.2. Statistical Analysis

#### 2.2.1. Uncertainty Analysis in Point Scale Simulations

For the first set of point scale simulations (i.e., those that used the observed local soil texture) and for the distributed simulations, we computed the uncertainties that PTF differences introduce in other variables as the average, across all time steps, of the standard deviation of modeled variables using different PTFs— that is, for a variable  $Y_i^h(t)$ , where *i* is the PTF, *h* the time scale of aggregation, and *t* the time step, the uncertainties are computed as:

$$\sigma_Y^h = \frac{1}{n_T} \sum_{t=1}^{n_T} \left( \sqrt{\frac{1}{n_P - 1} \sum_{i=1}^{n_P} \left( Y_i^h(t) - \overline{Y^h(t)} \right)^2} \right)$$
(1)

where  $n_T$  is the total number of time steps,  $n_P$  the number of PTFs and  $\overline{Y^h(t)} = \frac{1}{n_P} \sum_{i=1}^{n_P} Y_i^h(t)$ .

To obtain nondimensional coefficients of variation (CV) at the annual scale, the average value of annual standard deviations was normalized with the mean annual value of the corresponding flux for all PTFs. The variables chosen to describe water fluxes were the evapotranspiration (ET), plant transpiration (T), abiotic evaporation (E), comprising both ground evaporation and evaporation from interception, total runoff (R), and leakage to deeper soil layers (Lk), defined as the water flux at the bottom boundary of the simulation domain. To facilitate comparison of water fluxes across sites,  $\sigma_Y^h$  was normalized by the site's average precipitation at the same time scale. The variables chosen to describe carbon fluxes were the gross and net primary productivity (GPP and NPP), that is, the gross photosynthetic rate, and the net carbon used to produce plant tissues, that is, GPP minus autotrophic

respiration. The structure of vegetation was described by the leaf area index (LAI). The plant water stress was quantified by the vegetation stress factor  $\beta$ , which is the water stress reduction factor applied to photosynthesis when soil water potential is below a threshold.  $\beta$  ranges between 0 and 1, with 1 indicating unstressed vegetation and 0 complete suppression of photosynthesis.  $\beta$  is a sigmoidal function of soil water potential for values of water potential below a plant specific threshold (Wu et al., 2018). For the distributed simulations, the variation of LAI and ET for every grid point were also normalized by their average value across all time steps and PTFs for any given grid point, to obtain nondimensional coefficients of variation.

For the second set of simulations at the point scale (i.e., those that used 12 different soil classes), an approach similar to Paschalis et al. (2017) was used to partition variability into its natural component (i.e., stochastic variability) and its uncertainty due to different soil textures and different PTFs. For example, for a variable  $Y_{i,j,k}(t)$ , where *i* corresponds to the PTF, *j* to the soil texture, *k* to the site, and *t* is the time step, the natural variability is quantified by:

$$\sigma_{IAV} = \frac{1}{IJK} \sum_{i=1}^{I} \sum_{j=1}^{J} \sum_{k=1}^{K} \sqrt{\frac{1}{N-1} \sum_{i=1}^{N} \left( Y_{i,j,k}(t) - \frac{1}{N} \sum_{i=1}^{N} Y_{i,j,k}(t) \right)^2}$$
(2)

where *I*, *J*, *K*, *N* are the number of PTFs, soil textures, sites and years accordingly. Similarly, the PTF uncertainty is quantified by

$$\sigma_{PTF} = \frac{1}{JKN} \sum_{j=1}^{J} \sum_{k=1}^{K} \sum_{t=1}^{N} \sqrt{\frac{1}{I-1} \sum_{i=1}^{I} \left( Y_{i,j,k}(t) - \frac{1}{I} \sum_{i=1}^{I} Y_{i,j,k}(t) \right)^2}$$
(3)

And the texture uncertainty is:

$$\sigma_{texture} = \frac{1}{IKN} \sum_{i=1}^{I} \sum_{k=1}^{K} \sum_{t=1}^{N} \sqrt{\frac{1}{J-1} \sum_{i=1}^{J} \left( Y_{i,j,k}(t) - \frac{1}{J} \sum_{j=1}^{J} Y_{i,j,k}(t) \right)^2}$$
(4)

The variance partitioning was performed at the annual scale.

Uncertainty dependence on background climate was investigated using the wetness index of each site, defined as the average ratio of annual precipitation to annual potential evapotranspiration for all years in the meteorological

record 
$$\left(WI = \overline{\frac{P_y}{PET_Y}}\right)$$
.

#### 2.2.2. The Role of Topography

To understand how consideration of topography alters uncertainties, we computed the uncertainties introduced by PTFs for all grid points within the distributed domain, for ET, LAI, root zone soil moisture, and  $\beta$ . The dependence of the uncertainties on the topography was computed as a function of each grid point's topographic index, defined as  $TI = \log(A/\tan(b))$ , where  $A [m^2]$  is the area draining to any given grid point and b[rad] the slope at the same grid point defined as the steepest downhill slope based on the D8 flow direction method (e.g., Wilson et al., 2007). The catchment response was quantified using the probability distribution of discharge, expressed in terms of its survival function (i.e., probability of exceedance) at its outlet at the daily time scale.

#### 3. Results

#### 3.1. Plot Scale Simulations

The uncertainties introduced by the PTF disagreement in predicting soil hydraulic parameters are significant for all ecosystem variables. Considering annual scale statistics (Figure 3), uncertainty in vegetation structure is on average  $\overline{\sigma(LAI)} = 0.16\text{m}^2 \text{ m}^{-2}$  (Figure 3a) which is 8% of the LAI mean (Figure S3 in Supporting Information S1). The uncertainties in GPP and NPP are  $\overline{\sigma(GPP)} = 51.4 \text{ gC m}^{-2} \text{ y}^{-1}$  and  $\overline{\sigma(NPP)} = 30.2 \text{ gC m}^{-2} \text{ y}^{-1}$  (Figure 3b), which are 7.6% and 9.8% of their mean values, respectively (Figure S3 in Supporting Information S1).





**Figure 3.** Boxplots of the uncertainties for all 79 sites introduced by pedotransfer functions at the annual scale for (a) leaf area index, (b) gross and net primary productivity, and (c) water fluxes: evapotranspiration, plant transpiration, abiotic evaporation, runoff, and leakage.

The very similar magnitudes of LAI, GPP, and NPP uncertainties are expected as GPP directly depends on LAI, with photosynthetic rates scaling with LAI, and LAI in turn depends on GPP as it is controlled by the available carbon. As autotrophic respiration is less affected by plant water availability, the uncertainties of GPP and NPP are expected to be similar. GPP is affected by soil water availability (and thus PTF choice) in two ways: first, by short-term plant physiological responses that lead to stomatal closure and photosynthesis reduction, and second, by longer-term changes of LAI due to water availability.

Uncertainties introduced by PTFs in water fluxes at the annual scale were on average 4%, 3.4%, and 7.2% of the received precipitation, for evapotranspiration, runoff, and leakage, respectively (Figure 3c). Even though uncertainties are a small fraction of the total water received as precipitation at each site, when water fluxes are compared with their annual averages, uncertainties were on average 5.8%, 49.8%, and 169.8% of their mean values for evapotranspiration, leakage, and runoff, respectively (Figure S3 in Supporting Information S1). Runoff production is the most impacted variable in the experiments. Our results suggest that choosing among different PTFs can lead to large uncertainties with regard to runoff generation, without changing substantially the overall water budget. The reason for this behavior is that, in point scale experiments, simulated runoff is commonly a relatively small flux, as, by neglecting topography, we neglect flow convergence zones and thus the main locations for runoff generation. In addition, assuming a flat surface leads to infiltration reaching its maximum value, compared to steep hill slopes where infiltration rates may be lower.

The uncertainties of both hydrological and vegetation dynamics are dependent on the site's climate, albeit with a noticeable scatter (Figure 4). On average, uncertainties in evapotranspiration and plant productivity (NPP) increase in drier climates, the main reason being that, in arid/semiarid climates, vegetation is water stressed during part of the year, affecting both plant transpiration, which is the largest fraction of ET (Fatichi, Leuzinger, et al., 2016; Paschalis et al., 2018), and photosynthesis. In those climates, differences across PTFs lead to differences in the dynamics of soil water potential, thus affecting the strength and duration of plant water stress. In wetter climates, where water limitations occur rarely, uncertainties regarding productivity are small, despite considerable effects on soil water dynamics of the PTF choice. For climates where rainfall exceeds potential evapotranspiration (i.e., wetness index WI = P/PET > 1) uncertainties are mostly negligible, as plants do not experience significant water stress.

The uncertainties related to leakage (i.e., a proxy of groundwater recharge) are largest in intermediate climates (Figure 4). In arid/semiarid climates, irrespective of the PTF that describes the soil hydraulic conductivity and the shape of the characteristic curve, leakage is low due to persistent low soil moisture conditions. In mesic sites, where precipitation far exceeds potential evapotranspiration, uncertainties related to leakage are also small. In such sites, soil is commonly close to field capacity, and precipitation is mostly driving the amount of water





**Figure 4.** Dependence of the uncertainties of evapotranspiration, runoff, leakage, and net primary productivity on the wetness index. Uncertainties for water fluxes are expressed as the standard deviation of annual fluxes as computed for all pedotransfer functions, for every site (each dot), normalized by the average annual precipitation for each site. Solid lines correspond to a low-pass moving average filter.

recharge regardless of the soil type, with differences arising primarily due to PTF disagreement in the values of the saturated hydraulic conductivity. At intermediate sites, where soil moisture is fluctuating between dry and wet conditions, all uncertainties due to the saturated and unsaturated hydraulic conductivity as well as the shape of the characteristic curve can affect the simulated leakage. Runoff uncertainties are independent of climate, as they occur equally at all sites.

As many of the sites analyzed in this study experience a highly seasonal climate, uncertainty analysis was also performed at sub-seasonal scales, splitting seasons into dry and wet months, with dry months being defined as those when monthly potential evapotranspiration exceeds monthly rainfall and wet months the opposite. The uncertainty magnitudes and their dependence on climate are similar between dry and wet seasons (Figure S12 in Supporting Information S1), with the only exception being plant productivity. Uncertainty in plant productivity during dry seasons is on average lower than in the wet seasons, primarily in semiarid sites, because plant activity is at minimal rates no matter the choice of PTF.

#### 3.2. Soil Texture and PTF Choice

In Figure 5, we compare the total variability of both annual water and carbon fluxes due to the natural variability of climate (i.e., variability from 1 year to another due to random weather fluctuations) with uncertainties induced by soil texture and uncertainties due to the PTF choice. The interannual variability for NPP (and similarly GPP and LAI), ET (and similarly E and T) and Lk far exceeds the uncertainties due to either soil texture or the variability amongst PTFs (Figures 5a, 5b and 5d). This clearly illustrates the dominance of the model's forcing in a physically based model rather than its parameters in simulating these fluxes. What is of major importance, however, is that the uncertainties due to soil texture are comparable to, and in most cases lower than, the uncertainty in the simulated water and carbon fluxes than the uncertainties introduced by the choice of a given PTF once the soil texture is perfectly known. The within-soil type PTF variability for both water and carbon fluxes is mostly soil-type independent, with the exception of silt and silt-clay-loam experiencing the larger uncertainties for evapotranspiration and leakage (Figure S5 in Supporting Information S1) compared to the rest of the soil types used in this study.

Noticeably, the runoff uncertainties introduced by the stochastic variability of weather are similar in magnitude to the uncertainties introduced by soil texture and lower than PTF choice (Figure 5c). It should be highlighted,



## Water Resources Research





however, that a large part of this uncertainty relates to very low saturated hydraulic conductivity values predicted by the PTF of Weynants et al. (2009), leading to much higher values of runoff for this specific PTF. This disagreement is likely due to methodological differences in the quantification of saturated hydraulic conductivity. In the PTF of Weynants et al. (2009), the saturated hydraulic conductivity values used for PTF parameterization were computed using a parametric fit for conductivity values measured below saturation. This estimate eliminates the potential influence of macropore flow that would be present in all other PTFs, and for this reason, the estimates of conductivity of Weynants et al. (2009) are significantly lower (see also Figure 1a).

#### 3.3. Distributed Simulations

Uncertainties introduced by PTFs in both hydrological and vegetation dynamics can be either amplified or dampened when the effects of topography through radiation and lateral water fluxes are taken into account. As the patterns between GPP and NPP are very similar to LAI, only results pertaining to LAI are presented here. However, these are also representative of both gross and net primary productivity.

In the temperate ORNL site, where vegetation is not significantly water stressed, uncertainties in vegetation structure and ET at the annual scale are always <10% compared to their mean values (Figures 6a and 6b). LAI uncertainties always decline with increasing TI for both high and low permeability soils and are dampened when compared to the point scale simulations. However, large uncertainties occur in the small fraction of the catchment that experiences frequent inundation. The reason for this is that, in T&C, photosynthesis is suppressed for partially submerged or submerged vegetation, and small changes in the time a grid cell remains inundated can lead to large uncertainties in vegetation productivity and, ultimately, LAI. Also, radiation fluxes in (partially)





Figure 6. Scatter plots between the topographic index and the uncertainty introduced by different pedotransfer functions on simulated leaf area index (a-c) and evapotranspiration (b-d) at the annual scale for all grid points in the distributed domain. Upper row (a and b) corresponds to the ORNL site. Bottom row (c and d) corresponds to the Konza site. Blue colors correspond to simulations using a clayey soil, orange to simulations using a sandy soil.

submerged vegetation and vegetation dynamics are treated in a simplified manner in T&C. For this reason, we refrain from further interpreting this result, as it is dependent on the spatial resolution used for solving surface water fluxes and on the implementation of water-logging stress, which has not been fully tested in T&C. Additionally, as T&C adopts the simplified kinematic wave approximation for surface water dynamics and not the full dynamic wave approximation of the shallow water equations, inundation extent, which is the largest source of uncertainty for very high TI values, is unlikely to be well captured. Uncertainties regarding ET in ORNL are amplified, compared to the point scale simulations, for the low conductivity clayey soil, with the largest amplifications occurring in the flow convergence zones (high TI) (Figure 6b). In contrast, uncertainties are dampened when a sandy soil was used, with the dampening effect being strongest at intermediate TI values, which correspond to hillslopes near, but not on, the river network where water flow converges. Within the river network, a large scatter was obtained, primarily due to transpiration suppression due to inundation. For the same reasons as for LAI, we refrain from further interpretation of the uncertainties in channel areas.

In the water-limited Konza site, the uncertainties introduced by PTFs in simulated LAI values are large (up to 30% of the mean LAI values). Compared to the point scale simulation, uncertainties are amplified when a clay-rich soil was used and dampened in the case of a sandy soil (Figure 6c). ET patterns follow a similar trend (Figure 6d). Clay soils always exhibit higher uncertainties in the distributed simulations than sandy soils, even though for the equivalent plot scale simulations the differences were minimal (Figure 6). Uncertainties in the clay-soil simulation were always higher (for the entire simulation domain) than the equivalent point scale results for the water-limited Konza case, while for the sandy soils, uncertainties were dampened upland and amplified close to the flow convergence zones. Within the Konza catchment, uncertainties for both LAI and ET increase at high TIs close to the river network, with an exception being LAI uncertainties for TI < 9 in sandy soils.

In the water-limited site, most of the topographic variability of LAI can be explained by the uncertainties of simulated plant water stress (Figures 7c and 7d). The pattern for ET follows an almost identical pattern as for LAI (Figure S6 in Supporting Information S1), the reason being the dependence of transpiration, that constitutes the largest fraction of ET, on the amount of leaf area and thus NPP. Uncertainty in the stress factor  $\beta$  is not highly correlated with the uncertainties in soil moisture (Figures S7 and S8 in Supporting Information S1). The reason is that  $\beta$  depends on soil water potential in T&C and thus the uncertainties of soil water content are modified by the heavily nonlinear relationship between water content and soil water potential. Uncertainties in water potential are typically larger (Figure S2 in Supporting Information S1) at the drier end of the water retention curve, where plant water stress occurs. Overall, the soil hydraulic properties, the temporal structure of rainfall, and topography



### Water Resources Research



**Figure 7.** Scatter plot between the uncertainties in the simulated leaf area index for all the grid points in the distributed simulation domain as a function of the simulated uncertainties in the soil water stress factor ( $\beta$ ) at the annual scale. Colors indicate the topographic index. The upper row (a and b) corresponds to the ORNL site, the bottom row (c and d) to the Konza site. Left (a–c) and right (b–d) columns correspond to the clay and sandy soil simulations, respectively.

determine the water flow dynamics throughout the distributed domain and ultimately determine the duration and intensity of plant water stress in different regions of the catchment.

Conversely, in ORNL the impact of water stress on ET variability is mostly negligible, leading to lower total uncertainty values compared to the water-limited Konza site. In ORNL, both LAI and  $\beta$  uncertainties decline with the increase of TI for the sandy soil simulation, and are almost independent of TI for the clay soil simulation (Figure S9 in Supporting Information S1). The pattern of ET cannot be explained by  $\beta$  (Figures S8 in Supporting Information S1), suggesting that most of the variability comes from differences in net radiation and surface soil evaporation which is influenced by surface soil moisture. Surface soil moisture is simultaneously impacted by soil hydraulic properties and by radiation availability, modulated by topography due to hill slope aspect and shading. The overall complex interactions between these processes shape the dependence of ET uncertainties on local topography.

One variable that is highly impacted by the differences across PTFs is river discharge (Figure 8), and particularly its extremes. Uncertainties for extreme events can be within an order of magnitude and occur in both sites and for both soil types. In the highly permeable sandy soils, discharge, both in terms of base flow and extremes, is lower than in clay soils. Using the PTF of Weynants et al. (2009) leads to the highest simulated values of discharge. The PTF of Wösten et al. (1999) also leads to high discharge extremes. For the simulations using the low permeability clay soil the PTFs of Wösten et al. (1999); Zacharias and Wessolek (2007); Weynants et al. (2009) lead to the highest discharge extremes. As expected, in all cases, high runoff production was associated with PTFs having the lowest hydraulic conductivity values (Figure S10 in Supporting Information S1; first boxplot), which remarks that when the conductivity values are low, there is a higher than average discharge simulated.

All the results presented in this section are generalizable across time scales, from hourly to yearly, as the same patterns occur at all scales. However, its worth noticing that uncertainties overall decrease with an increasing temporal scale of aggregation (Figure S11 in Supporting Information S1).



**Figure 8.** Survival functions (exceedance probability) for the river discharge (R) at the catchment outlet for the distributed simulations using all seven pedotransfer functions. The upper row (a and b) corresponds to the Konza site. The bottom row (c and d) to the ORNL site. Left column (a–c) corresponds to the sandy soil simulation and right column (b–d) to the clay soil simulations.

#### 4. Discussion

#### 4.1. Impact of Differences Among PTFs

While many studies have focused on differences across PTFs and their ability of accurately predicting soil hydraulic properties (e.g., Minasny et al., 1999), how uncertainties in soil hydraulic properties impact ecosystem dynamics at various spatial scales has received far less attention. Our results show that both hydrological and vegetation dynamics can be substantially affected by this source of uncertainty and that PTF choice might be as important, or generally more, than the choice of the soil type, as already found by Gutmann and Small (2007). Previous studies focused primarily on hydrological dynamics and our results are in agreement with their findings. Weihermüller et al. (2021) showed that differences across PTFs can lead to a coefficient of variation for ET of  $\sim 5\%$ , in line with our 5.8% estimate, despite the very different modeling approaches used. Interestingly, Weihermüller et al. (2021) report a large number of cases where some PTFs led to numerical instabilities in the HYDRUS model. This problem did not occur in the T&C simulations, partly because of the different numerical scheme (finite volumes vs. finite elements) and partly because of the correction of Webb (2000) we introduced in the MvG model to improve representation of the water retention curves in the driest regime. Our results were also in line with Edouard et al. (2018); Yang et al. (2018); Rieger et al. (2010); Mohajerani et al. (2021) showing large uncertainties regarding runoff production, even though we cannot directly compare the uncertainties due to differences in climates and topography with the case studies presented in previous research. However, a crucial result is that uncertainties in runoff generation are of the same order of magnitude in all climates and soil types analyzed (Figure 4). As knowledge of precise hydraulic properties is not common in most parts of the world, this finding highlights the long-standing challenge in hydrology of making flood risk simulations based solely on a process-based approach, which entails irreducible uncertainties.

Beyond runoff generation, the hydrological flux mostly affected by uncertainty was water percolation to deeper soil layers (i.e., leakage), which is a proxy of groundwater recharge. Uncertainties regarding leakage were largest, as a fraction of precipitation, at locations of intermediate wetness. In such areas, pumped groundwater is commonly a large fraction of the total water use (Gleeson et al., 2012), highlighting the importance of the uncertainty propagation from PTF estimates to recharge dynamics. The total uncertainties could also be larger as PTFs are mostly trained on surface soil data (e.g., Rawls et al., 1991). To better tackle this problem, more data are needed regarding soil hydraulic properties of deeper layers to further parameterize PTFs.

#### 4.2. Implications for Carbon Cycle and Terrestrial Biosphere Models

This study is the first to estimate how uncertainties introduced by PTF choice impact carbon cycle dynamics. Our results suggest that, on average, variability of soil hydraulic parameters can lead to ~10% uncertainty in both carbon fluxes and vegetation structure (i.e., LAI). This uncertainty, even though large, is much smaller than the uncertainties reported between different terrestrial biosphere models (typically in the order of 10%–40% for GPP and NPP for different parts of the world; e.g., Keenan & Williams, 2018; Zheng et al., 2020; Cramer et al., 1999). Combined uncertainties related to differences in how they simulate processes from photosynthesis (e.g., Pappas et al., 2013; Rogers, 2014), to phenology (e.g., De Kauwe et al., 2017; Richardson et al., 2013), carbon allocation (e.g., Fatichi, Pappas, et al., 2019; Franklin et al., 2012), and vegetation water stress (e.g., Wu et al., 2018; Paschalis et al., 2020) largely exceed the estimated uncertainties introduced by PTFs. However, an uncertainty for LAI and carbon fluxes in the order of ~10% is still significant and comparable to changes expected with climate change (e.g., Shao et al., 2013). Overall, reducing epistemic uncertainties between current generation terrestrial biosphere models and the uncertainty introduced by their most sensitive parameters is still of high priority for better carbon cycle predictions (e.g., Lovenduski & Bonan, 2017; Pappas et al., 2013, 2016).

Even though PTF uncertainties introduced in LAI and carbon fluxes are less than other sources, their importance is still high, particularly in water-limited ecosystems. Such ecosystems are amongst the most vulnerable under a changing climate (e.g., Huang et al., 2017), and projected increases in atmospheric, meteorological and hydrologic drought can intensify plant water stress. Changes in plant water stress can alter both the local water and carbon cycles, with pronounced impacts on global scale carbon dynamics (e.g., Humphrey et al., 2018) and also on the livelihood of people, as in many semiarid areas globally, subsistence is to a large degree dependent on rainfed agriculture, particularly in the developing world. It is fundamental to remark that a proper quantification of plant water stress cannot ignore a robust definition of soil hydraulic properties. While a large number of recent studies have focused on the importance of representing plant hydraulic and physiological properties in simulating water stress in terrestrial biosphere models (e.g., Feng, 2020; Kennedy et al., 2019; Medlyn et al., 2016; Xu et al., 2016), much less attention has been dedicated to soil hydraulic properties, which might be equally important in defining plant-water stress.

PTFs-induced uncertainties in plant water stress were able to explain, to a large extent, the uncertainties in both water and carbon dynamics. However, uncertainties in plant water stress were not highly correlated with uncertainties in soil moisture (Figure S8 in Supporting Information S1). The reason is that, in the model we used, plant water stress depends on root integrated soil water potential (which is the quantity felt by plants) and not soil moisture. Many alternative models use soil moisture for parameterizing plant water stress (e.g., Harper et al., 2021; Paschalis et al., 2020; Wu et al., 2018), and thus the uncertainties of both the water and carbon dynamics, particularly in drier areas, are expected to be less PTF-dependent and more dependent on how soil moisture is used to parameterize water stress. This highlights the importance of simultaneously monitoring both soil water content and vegetation stress dynamics, in order to reduce uncertainties and, at the same time, improve and best tune terrestrial biosphere models. New generation remote sensing technologies can now provide to some degree such information at the global scale (e.g., Chan et al., 2016; Konings et al., 2021; Y. Sun et al., 2015).

#### 4.3. The Importance of Accounting for Topographic Effects

Small scale topography was found to both amplify and dampen (based on soil type and climate) uncertainties introduced by the choice of the PTF. Particularly, in dry conditions, uncertainties within the spatially distributed simulations were up to threefold larger than plot scale simulations, for both water and carbon dynamics. This effect is largely related to lateral water redistribution in the surface (runoff and runon processes) and subsurface, which is considerably affected by soil hydraulic properties. In a distributed domain, such effects are accumulated from upland areas toward down slope converging areas, considerably increasing the importance of soil hydraulic properties in the simulation of plant water stress and thus carbon and water fluxes. To a second order, topographic variability also implies a different amount of received net radiation at the surface, which interacts with ground evaporation. Such a process is also mediated by soil hydraulic properties. Currently, all global scale applications of ecosystem models, either offline or coupled with atmospheric models, neglect subgrid topography and lateral water fluxes. This potentially leads to an underestimation of the importance of properly representing soil hydraulic properties (Fatichi et al., 2020; Tafasca et al., 2020). This might have implications for a number of dynamics, such as land surface coupling (Gevaert et al., 2018), runoff predictions (e.g., Ukkola et al., 2018; X. Zhang

et al., 2014) and predictions of plant water stress and its impacts on the carbon cycle (e.g., Anderegg et al., 2013; De Kauwe et al., 2020). Introduction of new generation hyper-resolution models can potentially resolve these problems, albeit with large computational costs involved (e.g., Fan et al., 2019; Forrester & Maxwell, 2020; Mastrotheodoros et al., 2020; O'Neill et al., 2020).

One additional source of uncertainty related to soil hydraulic properties, not investigated here, is the role of soil structure (Bonetti et al., 2021; Fatichi et al., 2020). All PTFs used in this study neglect the effect of soil structure that can significantly alter the soil hydraulic parameters at water potentials close to saturation. In fact, the uncertainties expected from soil structure alone have been found to have a similar order of magnitude as the one reported in this study, particularly for hydrological dynamics (Fatichi et al., 2020). Ultimately, uncertainties introduced by soil structure are further increased by complex soil-vegetation combinations (Bonetti et al., 2021), and can significantly impact hydrological and land surface dynamics. Based on results obtained here, it is likely that the role of soil structure in modifying ecohydrological variables will be magnified by considering a distributed topography, as already hinted by Fatichi et al. (2020).

#### 4.4. Study Limitations

The results of the present study were derived using a single ecohydrological model. One source of uncertainty that was not considered here is the role of the model structure itself. The different estimates of soil hydraulic properties across PTFs should affect soil water transport similarly amongst most ecohydrological and terrestrial biosphere models, all other things being equal. That is because most models solve the Richards' equation (albeit exceptions that use simplified hydrological dynamics exist; see Wu et al. (2018) for a review of how different models solve soil water dynamics), commonly using similar numerical procedures. However, the representation of how soil moisture impacts plant water stress and thus plant transpiration and productivity is highly variable across models (e.g., Egea et al., 2011; Paschalis et al., 2020; Wu et al., 2018; Zhou et al., 2013). These model-specific differences will impact all ecohydrological dynamics. Similar large differences across models can be expected for the parametrization of the dependence of soil evaporation on surface soil moisture (e.g., Lehmann et al., 2018, 2008) and also how plants access water in the root zone (e.g., Harper et al., 2021). To compute the magnitude of uncertainties related to the choice of the ecosystem model, a multimodel intercomparison project would be necessary, or even better, an analysis where single model components are replaced with alternative solutions within the same modeling framework.

In this study, we analyzed how uncertainties of soil hydraulic properties propagate to hydrological and ecosystem dynamics. However, disentangling the exact mechanism of how those uncertainties propagate across spatial and temporal scales is a major challenge. Many of the variables we analyzed are highly correlated (Figure S13 in Supporting Information S1) and causally dependent. For example, uncertainties regarding plant water stress will, at short time scales, impact plant transpiration and GPP/NPP through stomata closure. At longer time scales, LAI is reduced if GPP decreases, and it can subsequently impact both water and carbon fluxes. To disentangle the exact mechanisms, multiscale causal inference techniques would be needed (e.g., Detto et al., 2012; Runge et al., 2019).

Finally, we used for all sites and soil texture combinations all PTFs, without filtering out potentially unrealistic results, as the scope of the study was to compute the uncertainties originated by the indiscriminate use of PTFs, beyond specific areas and limits of derivation as often done in large-scale models. Developing a filtering scheme that could identify the applicability range of each PTF would likely reduce their uncertainties. In this sense, our estimates are likely conservative and quantify the maximum values of the PTF related uncertainties. Approaches that can successfully reduce uncertainties related to PTFs have been previously applied (e.g., McNeall et al., 2016; Verhoef & Egea, 2014), and could be incorporated in global scale modeling applications where currently PTFs are used without any filtering.

#### 5. Conclusions

In this work, we quantified how disagreement between predicted hydraulic properties by 7 common pedotransfer functions leads to uncertainties in the simulated carbon and water fluxes at 79 sites across the world. We found that PTF disagreement can significantly impact both the water and carbon cycles, irrespective of soil type. Runoff and leakage to deep soil layers were the two hydrological fluxes most impacted by this uncertainty, with the



uncertainty of annual fluxes being on average 168% and 50% compared to their mean values. Runoff uncertainties were independent of climate, while leakage uncertainties were strongest at climates of intermediate wetness (i.e., a wetness index of around 1). Despite those large uncertainties, because of the relatively small magnitude of those fluxes compared to annual rainfall, the uncertainties they introduce on the partitioning of precipitation to evapotranspiration and the other hydrological budget components is within 10%. Uncertainties in leaf area index and carbon fluxes (GPP and NPP), computed as the annual coefficient of variation, were smaller (8%, 7.6%, and 9.8%) than water fluxes and to a large extent explained by the intensity and duration of plant water stress. For both hydrological and vegetation dynamics, the uncertainties due to PTF disagreements were larger than the uncertainties due to differences in soil types, highlighting the importance of the choice of a PTF and the need for new, more robust and generally applicable PTFs. Finally, the results showed that small scale topography can amplify the importance of the uncertainties of soil hydraulic parameterization by threefold, especially for low permeability, clay-rich soils in water-limited ecosystems, highlighting that new generation hyper-resolution and distributed climate and terrestrial biosphere models might be more affected by soil hydraulic parameters than the current ones.

#### **Data Availability Statement**

The model used for the simulations can be found in the following citable repository, https://doi.org/10.24433/ CO.0905087.v2.

#### References

- Anderegg, L. D., Anderegg, W. R., & Berry, J. A. (2013). Not all droughts are created equal: Translating meteorological drought into woody plant mortality. *Tree Physiology*, 33(7), 672–683. https://doi.org/10.1093/treephys/tpt044
- Assouline, S., & Or, D. (2006). Anisotropy factor of saturated and unsaturated soils. *Water Resources Research*, 42(12). https://doi.org/10.1029/2006WR005001
- Assouline, S., & Or, D. (2013). Conceptual and parametric representation of soil hydraulic properties: A review. Vadose Zone Journal, 12(4), vzj201307. https://doi.org/10.2136/vzj2013.07.0121
- Bonan, G. (2019). Climate change and terrestrial ecosystem modeling. Cambridge University Press. https://doi.org/10.1017/9781107339217
  - Bonetti, S., Wei, Z., & Or, D. (2021). A framework for quantifying hydrologic effects of soil structure across scales. *Communications Earth & Environment*, 2(1), 1–10. https://doi.org/10.1038/s43247-021-00180-0
  - Brooks, R. H., & Corey, A. T. (1964). Hydraulic properties of porous media. Colorado State University.
  - Chan, S. K., Bindlish, R., O'Neill, P. E., Njoku, E., Jackson, T., Colliander, A., et al. (2016). Assessment of the SMAP passive soil moisture product. *IEEE Transactions on Geoscience and Remote Sensing*, 54(8), 4994–5007. https://doi.org/10.1109/TGRS.2016.2561938
  - Chirico, G., Medina, H., & Romano, N. (2010). Functional evaluation of PTF prediction uncertainty: An application at hillslope scale. *Geoderma*, 155(3–4), 193–202. https://doi.org/10.1016/j.geoderma.2009.06.008
  - Collatz, G. J., Ball, J. T., Grivet, C., & Berry, J. A. (1991). Physiological and environmental regulation of stomatal conductance, photosynthesis and transpiration: A model that includes a laminar boundary layer. Agricultural and Forest Meteorology, 54(2–4), 107–136. https://doi. org/10.1016/0168-1923(91)90002-8
  - Collatz, G. J., Ribas-Carbo, M., & Berry, J. (1992). Coupled photosynthesis-stomatal conductance model for leaves of C4 plants. Functional Plant Biology, 19(5), 519–538. https://doi.org/10.1071/PP9920519
  - Cooper, E., Blyth, E., Cooper, H., Ellis, R., Pinnington, E., & Dadson, S. J. (2020). Using data assimilation to optimize pedotransfer functions using large-scale in-situ soil moisture observations. *Hydrology and Earth System Sciences*, 25(2), 2445–2458.
  - Cosby, B., Hornberger, G., Clapp, R., & Ginn, T. (1984). A statistical exploration of the relationships of soil moisture characteristics to the physical properties of soils. Water Resources Research, 20(6), 682–690. https://doi.org/10.1029/WR020i006p00682
  - Cramer, W., Kicklighter, D. W., Bondeau, A., Iii, B. M., Churkina, G., Nemry, B., et al. (1999). Comparing global models of terrestrial net primary productivity (NPP): Overview and key results. *Global Change Biology*, 5(S1), 1–15. https://doi.org/10.1046/j.1365-2486.1999.00009.x
  - Dai, Y., Shangguan, W., Wei, N., Xin, Q., Yuan, H., Zhang, S., et al. (2019). A review of the global soil property maps for Earth system models. Soils, 5(2), 137–158. https://doi.org/10.5194/soil-5-137-2019
  - De Kauwe, M. G., Medlyn, B. E., Ukkola, A. M., Mu, M., Sabot, M. E., Pitman, A. J., et al. (2020). Identifying areas at risk of drought-induced tree mortality across South-Eastern Australia. *Global Change Biology*, 26(10), 5716–5733. https://doi.org/10.1111/gcb.15215
  - De Kauwe, M. G., Medlyn, B. E., Walker, A. P., Zaehle, S., Asao, S., Guenet, B., et al. (2017). Challenging terrestrial biosphere models with data from the long-term multifactor prairie heating and CO<sub>2</sub> enrichment experiment. *Global Change Biology*, 23(9), 3623–3645. https://doi. org/10.1111/gcb.13643
  - Deng, H., Ye, M., Schaap, M. G., & Khaleel, R. (2009). Quantification of uncertainty in pedotransfer function-based parameter estimation for unsaturated flow modeling. *Water Resources Research*, 45(4). https://doi.org/10.1029/2008WR007477
  - Detto, M., Molini, A., Katul, G., Stoy, P., Palmroth, S., & Baldocchi, D. (2012). Causality and persistence in ecological systems: A nonparametric spectral granger causality approach. *The American Naturalist*, 179(4), 524–535. https://doi.org/10.1086/664628
  - Edouard, S., Vincendon, B., & Ducrocq, V. (2018). Ensemble-based flash-flood modelling: Taking into account hydrodynamic parameters and initial soil moisture uncertainties. *Journal of Hydrology*, 560, 480–494. https://doi.org/10.1016/j.jhydrol.2017.04.048
  - Egea, G., Verhoef, A., & Vidale, P. L. (2011). Towards an improved and more flexible representation of water stress in coupled photosynthesis– stomatal conductance models. Agricultural and Forest Meteorology, 151(10), 1370–1384. https://doi.org/10.1016/j.agrformet.2011.05.019
  - Fan, Y., Clark, M., Lawrence, D. M., Swenson, S., Band, L., Brantley, S. L., et al. (2019). Hillslope hydrology in global change research and Earth system modeling. *Water Resources Research*, 55(2), 1737–1772. https://doi.org/10.1029/2018WR023903

#### Acknowledgments

AP acknowledges funding from NERC (NE/S003495/1). SF would like to thank the support of the National University of Singapore through the project "Bridging scales from below: The role of heterogeneities in the global water and carbon budgets", Award Number: 22-3637-A0001.

- Farthing, M. W., & Ogden, F. L. (2017). Numerical solution of Richards' equation: A review of advances and challenges. Soil Science Society of America Journal, 81(6), 1257–1269. https://doi.org/10.2136/sssaj2017.02.0058
- Fatichi, S., Ivanov, V. Y., & Caporali, E. (2012). A mechanistic ecohydrological model to investigate complex interactions in cold and warm water-controlled environments: 1. Theoretical framework and plot-scale analysis. *Journal of Advances in Modeling Earth Systems*, 4(2). https://doi.org/10.1029/2011MS000086
- Fatichi, S., Leuzinger, S., Paschalis, A., Langley, J. A., Barraclough, A. D., & Hovenden, M. J. (2016). Partitioning direct and indirect effects reveals the response of water-limited ecosystems to elevated CO<sub>2</sub>. Proceedings of the National Academy of Sciences, 113(45), 12757–12762. https://doi.org/10.1073/pnas.1605036113
- Fatichi, S., Manzoni, S., Or, D., & Paschalis, A. (2019). A mechanistic model of microbially mediated soil biogeochemical processes: A reality check. *Global Biogeochemical Cycles*, 33(6), 620–648. https://doi.org/10.1029/2018GB006077
- Fatichi, S., Or, D., Walko, R., Vereecken, H., Young, M. H., Ghezzehei, T. A., et al. (2020). Soil structure is an important omission in Earth system models. *Nature Communications*, 11(1), 1–11. https://doi.org/10.1038/s41467-020-14411-z
- Fatichi, S., & Pappas, C. (2017). Constrained variability of modeled I: ET ratio across biomes. *Geophysical Research Letters*, 44(13), 6795–6803. https://doi.org/10.1002/2017GL074041
- Fatichi, S., Pappas, C., Zscheischler, J., & Leuzinger, S. (2019). Modelling carbon sources and sinks in terrestrial vegetation. New Phytologist, 221(2), 652–668. https://doi.org/10.1111/nph.15451
- Fatichi, S., Peleg, N., Mastrotheodoros, T., Pappas, C., & Manoli, G. (2021). An ecohydrological journey of 4500 years reveals a stable but threatened precipitation–groundwater recharge relation around Jerusalem. *Science Advances*, 7(37), eabe6303. https://doi.org/10.1126/sciadv. abe6303
- Fatichi, S., Vivoni, E. R., Ogden, F. L., Ivanov, V. Y., Mirus, B., Gochis, D., et al. (2016). An overview of current applications, challenges, and future trends in distributed process-based models in hydrology. *Journal of Hydrology*, 537, 45–60. https://doi.org/10.1016/j.jhydrol.2016.03.026
- Feng, X. (2020). Marching in step: The importance of matching model complexity to data availability in terrestrial biosphere models. *Global Change Biology*, 26(6), 3190–3192. https://doi.org/10.1111/gcb.15090
- Fisher, J. B., Huntzinger, D. N., Schwalm, C. R., & Sitch, S. (2014). Modeling the terrestrial biosphere. Annual Review of Environment and Resources, 39, 91–123. https://doi.org/10.1146/annurev-environ-012913-093456
- Forrester, M. M., & Maxwell, R. M. (2020). Impact of lateral groundwater flow and subsurface lower boundary conditions on atmospheric boundary layer development over complex terrain. *Journal of Hydrometeorology*, 21(6), 1133–1160. https://doi.org/10.1175/JHM-D-19-0029.1
- Franklin, O., Johansson, J., Dewar, R. C., Dieckmann, U., McMurtrie, R. E., Brännström, Å., & Dybzinski, R. (2012). Modeling carbon allocation in trees: A search for principles. *Tree Physiology*, 32(6), 648–666. https://doi.org/10.1093/treephys/tpr138
- Gevaert, A., Miralles, D. G., de Jeu, R. A., Schellekens, J., & Dolman, A. J. (2018). Soil moisture-temperature coupling in a set of land surface models. *Journal of Geophysical Research: Atmospheres*, 123(3), 1481–1498. https://doi.org/10.1002/2017JD027346
- Gleeson, T., Wada, Y., Bierkens, M. F., & Van Beek, L. P. (2012). Water balance of global aquifers revealed by groundwater footprint. *Nature*, 488(7410), 197–200. https://doi.org/10.1038/nature11295
- Gupta, S., Hengl, T., Lehmann, P., Bonetti, S., & Or, D. (2020). SoilKsatDB: Global soil saturated hydraulic conductivity measurements for geoscience applications. *Earth System Science Data Discussions*, 1–26. https://doi.org/10.5194/essd-13-1593-2021
- Gutmann, E. D., & Small, E. E. (2007). A comparison of land surface model soil hydraulic properties estimated by inverse modeling and pedotransfer functions. *Water Resources Research*, 43(5). https://doi.org/10.1029/2006WR005135
- Harper, A. B., Williams, K. E., McGuire, P. C., Duran Rojas, M. C., Hemming, D., Verhoef, A., et al. (2021). Improvement of modeling plant responses to low soil moisture in JULESvn4. 9 and evaluation against flux tower measurements. *Geoscientific Model Development*, 14(6), 3269–3294. https://doi.org/10.5194/gmd-14-3269-2021
- Hopp, L., Fatichi, S., & Ivanov, V. Y. (2016). Simulating water flow in variably saturated soils: A comparison of a 3-D model with approximationbased formulations. *Hydrology Research*, 47(2), 274–290. https://doi.org/10.2166/nh.2015.126
- Huang, J., Li, Y., Fu, C., Chen, F., Fu, Q., Dai, A., et al. (2017). Dryland climate change: Recent progress and challenges. *Reviews of Geophysics*, 55(3), 719–778. https://doi.org/10.1002/2016RG000550
- Humphrey, V., Zscheischler, J., Ciais, P., Gudmundsson, L., Sitch, S., & Seneviratne, S. I. (2018). Sensitivity of atmospheric CO<sub>2</sub> growth rate to observed changes in terrestrial water storage. *Nature*, 560(7720), 628–631. https://doi.org/10.1038/s41586-018-0424-4
- Ivanov, V. Y., Bras, R. L., & Vivoni, E. R. (2008). Vegetation-hydrology dynamics in complex terrain of semiarid areas: 2. Energy-water controls of vegetation spatiotemporal dynamics and topographic niches of favorability. Water Resources Research, 44(3). https://doi. org/10.1029/2006WR005595
- Keenan, T., & Williams, C. (2018). The terrestrial carbon sink. Annual Review of Environment and Resources, 43, 219–243. https://doi.org/10.1146/annurev-environ-102017-030204
- Kennedy, D., Swenson, S., Oleson, K. W., Lawrence, D. M., Fisher, R., Lola da Costa, A. C., & Gentine, P. (2019). Implementing plant hydraulics in the community land model, version 5. *Journal of Advances in Modeling Earth Systems*, 11(2), 485–513. https://doi.org/10.1029/2018MS001500 Konings, A. G., Holtzman, N. M., Rao, K., Xu, L., & Saatchi, S. S. (2021). Interannual variations of vegetation optical depth are due to both water
- stress and biomass changes. *Geophysical Research Letters*, 48(16), e2021). Interaindar variations do vogetation optical deput at due to both water stress and biomass changes. *Geophysical Research Letters*, 48(16), e2021GL095267. https://doi.org/10.1029/2021GL095267
- Lehmann, P., Assouline, S., & Or, D. (2008). Characteristic lengths affecting evaporative drying of porous media. *Physical Review E*, 77(5), 056309. https://doi.org/10.1103/PhysRevE.77.056309
- Lehmann, P., Merlin, O., Gentine, P., & Or, D. (2018). Soil texture effects on surface resistance to bare-soil evaporation. *Geophysical Research Letters*, 45(19), 10–398. https://doi.org/10.1029/2018GL078803
- Loosvelt, L., Pauwels, V. R., Cornelis, W. M., De Lannoy, G. J., & Verhoest, N. E. (2011). Impact of soil hydraulic parameter uncertainty on soil moisture modeling. *Water Resources Research*, 47(3). https://doi.org/10.1029/2010WR009204
- Lovenduski, N. S., & Bonan, G. B. (2017). Reducing uncertainty in projections of terrestrial carbon uptake. *Environmental Research Letters*, 12(4), 044020. https://doi.org/10.1088/1748-9326/aa66b8
- Mastrotheodoros, T., Pappas, C., Molnar, P., Burlando, P., Manoli, G., Parajka, J., et al. (2020). More green and less blue water in the Alps during warmer summers. *Nature Climate Change*, 10(2), 155–161. https://doi.org/10.1038/s41558-019-0676-5
- McNeall, D., Williams, J., Booth, B., Betts, R., Challenor, P., Wiltshire, A., & Sexton, D. (2016). The impact of structural error on parameter constraint in a climate model. *Earth System Dynamics*, 7(4), 917–935. https://doi.org/10.5194/esd-7-917-2016
- Medlyn, B. E., De Kauwe, M. G., & Duursma, R. A. (2016). New developments in the effort to model ecosystems under water stress. New Phytologist, 212(1), 5–7. https://doi.org/10.1111/nph.14082
- Minasny, B., McBratney, A. B., & Bristow, K. L. (1999). Comparison of different approaches to the development of pedotransfer functions for water-retention curves. *Geoderma*, 93(3–4), 225–253. https://doi.org/10.1016/S0016-7061(99)00061-0

- Mohajerani, H., Teschemacher, S., & Casper, M. C. (2021). A comparative investigation of various pedotransfer functions and their impact on hydrological simulations. *Water*, *13*(10), 1401. https://doi.org/10.3390/w13101401
- Moustakis, Y., Fatichi, S., Onof, C., & Paschalis, A. (2022). Insensitivity of ecosystem productivity to predicted changes in fine-scale rainfall variability. *Journal of Geophysical Research: Biogeosciences*, 127(2), e2021JG006735. https://doi.org/10.1029/2021JG006735
- Mualem, Y. (1976). A new model for predicting the hydraulic conductivity of unsaturated porous media. *Water Resources Research*, *12*(3), 513–522. https://doi.org/10.1029/WR012i003p00513
- O'Neill, M. M., Tijerina, D. T., Condon, L. E., & Maxwell, R. M. (2020). Assessment of the ParFlow-CLM CONUS 1.0 integrated hydrologic model: Evaluation of hyper-resolution water balance components across the contiguous United States. *Geoscientific Model Development Discussions*, 1–50. https://doi.org/10.5194/gmd-14-7223-2021
- Pappas, C., Fatichi, S., & Burlando, P. (2016). Modeling terrestrial carbon and water dynamics across climatic gradients: Does plant trait diversity matter? New Phytologist, 209(1), 137–151. https://doi.org/10.1111/nph.13590
- Pappas, C., Fatichi, S., Leuzinger, S., Wolf, A., & Burlando, P. (2013). Sensitivity analysis of a process-based ecosystem model: Pinpointing parameterization and structural issues. *Journal of Geophysical Research: Biogeosciences*, 118(2), 505–528. https://doi.org/10.1002/jgrg.20035Paschalis, A., Fatichi, S., Katul, G. G., & Ivanov, V. Y. (2015). Cross-scale impact of climate temporal variability on ecosystem water and carbon
- fluxes. Journal of Geophysical Research: Biogeosciences, 120(9), 1716–1740. https://doi.org/10.1002/2015JG003002
- Paschalis, A., Fatichi, S., Pappas, C., & Or, D. (2018). Covariation of vegetation and climate constrains present and future t/et variability. *Environmental Research Letters*, 13(10), 104012. https://doi.org/10.1088/1748-9326/aae267
- Paschalis, A., Fatichi, S., Zscheischler, J., Ciais, P., Bahn, M., Boysen, L., et al. (2020). Rainfall manipulation experiments as simulated by terrestrial biosphere models: Where do we stand? *Global Change Biology*, 26(6), 3336–3355. https://doi.org/10.1111/gcb.15024
- Paschalis, A., Katul, G. G., Fatichi, S., Palmroth, S., & Way, D. (2017). On the variability of the ecosystem response to elevated atmospheric CO<sub>2</sub> across spatial and temporal scales at the duke forest face experiment. Agricultural and Forest Meteorology, 232, 367–383. https://doi. org/10.1016/j.agrformet.2016.09.003
- Patil, N. G., & Singh, S. K. (2016). Pedotransfer functions for estimating soil hydraulic properties: A review. Pedosphere, 26(4), 417–430. https:// doi.org/10.1016/S1002-0160(15)60054-6
- Pinnington, E., Amezcua, J., Cooper, E., Dadson, S., Ellis, R., Peng, J., et al. (2021). Improving soil moisture prediction of a high-resolution land surface model by parameterising pedotransfer functions through assimilation of SMAP satellite data. *Hydrology and Earth System Sciences*, 25(3), 1617–1641. https://doi.org/10.5194/hess-25-1617-2021
- Rawls, W., Gish, T., & Brakensiek, D. (1991). Estimating soil water retention from soil physical properties and characteristics. In Advances in soil science (pp. 213–234). Springer. https://doi.org/10.1007/978-1-4612-3144-8\_5
- Remondi, F., Botter, M., Burlando, P., & Fatichi, S. (2019). Variability of transit time distributions with climate and topography: A modelling approach. Journal of Hydrology, 569, 37–50. https://doi.org/10.1016/j.jhydrol.2018.11.011
- Richardson, A. D., Keenan, T. F., Migliavacca, M., Ryu, Y., Sonnentag, O., & Toomey, M. (2013). Climate change, phenology, and phenological control of vegetation feedbacks to the climate system. Agricultural and Forest Meteorology, 169, 156–173. https://doi.org/10.1016/j. agrformet.2012.09.012
- Rieger, W., Winter, F., & Disse, M. (2010). Uncertainties of soil parameterisation in process-based simulation of distributed flood control measures. Advances in Geosciences, 27, 121–129. https://doi.org/10.5194/adgeo-27-121-2010
- Rogers, A. (2014). The use and misuse of V c, max in Earth System Models. Photosynthesis Research, 119(1), 15-29. https://doi.org/10.1007/s11120-013-9818-1
- Runge, J., Bathiany, S., Bollt, E., Camps-Valls, G., Coumou, D., Deyle, E., et al. (2019). Inferring causation from time series in Earth system sciences. *Nature Communications*, 10(1), 1–13. https://doi.org/10.1038/s41467-019-10105-3
- Saxton, K., Rawls, W., Romberger, J. S., & Papendick, R. (1986). Estimating generalized soil-water characteristics from texture. Soil Science Society of America Journal, 50, 1031. https://doi.org/10.2136/sssaj1986.03615995005000040054x
- Saxton, K. E., & Rawls, W. J. (2006). Soil water characteristic estimates by texture and organic matter for hydrologic solutions. Soil Science Society of America Journal, 70(5), 1569–1578. https://doi.org/10.2136/sssaj2005.0117
- Schwalm, C. R., Schaefer, K., Fisher, J. B., Huntzinger, D., Elshorbany, Y., Fang, Y., et al. (2019). Divergence in land surface modeling: Linking spread to structure. *Environmental Research Communications*, 1(11), 111004. https://doi.org/10.1088/2515-7620/ab4a8a
- Shahraeeni, E., Lehmann, P., & Or, D. (2012). Coupling of evaporative fluxes from drying porous surfaces with air boundary layer: Characteristics of evaporation from discrete pores. Water Resources Research, 48(9). https://doi.org/10.1029/2012WR011857
- Shao, P., Zeng, X., Sakaguchi, K., Monson, R. K., & Zeng, X. (2013). Terrestrial carbon cycle: Climate relations in eight CMIP5 Earth system models. *Journal of Climate*, 26(22), 8744–8764. https://doi.org/10.1175/JCLI-D-12-00831.1
- Sun, W., Yao, X., Cao, N., Xu, Z., & Yu, J. (2016). Integration of soil hydraulic characteristics derived from pedotransfer functions into hydrological models: Evaluation of its effects on simulation uncertainty. *Hydrology Research*, 47(5), 964–978. https://doi.org/10.2166/nh.2016.150
- Sun, Y., Fu, R., Dickinson, R., Joiner, J., Frankenberg, C., Gu, L., et al. (2015). Drought onset mechanisms revealed by satellite solar-induced chlorophyll fluorescence: Insights from two contrasting extreme events. *Journal of Geophysical Research: Biogeosciences*, 120(11), 2427– 2440. https://doi.org/10.1002/2015JG003150
- Tafasca, S., Ducharne, A., & Valentin, C. (2020). Weak sensitivity of the terrestrial water budget to global soil texture maps in the ORCHIDEE land surface model. *Hydrology and Earth System Sciences*, 24(7), 3753–3774. https://doi.org/10.5194/hess-24-3753-2020
- Tuzet, A., Perrier, A., & Leuning, R. (2003). A coupled model of stomatal conductance, photosynthesis and transpiration. *Plant, Cell and Environment*, 26(7), 1097–1116. https://doi.org/10.1046/j.1365-3040.2003.01035.x
- Ukkola, A., Pitman, A., De Kauwe, M., Abramowitz, G., Herger, N., Evans, J., & Decker, M. (2018). Evaluating CMIP5 model agreement for multiple drought metrics. *Journal of Hydrometeorology*, 19(6), 969–988. https://doi.org/10.1175/JHM-D-17-0099.1
- Van Genuchten, M. T. (1980). A closed-form equation for predicting the hydraulic conductivity of unsaturated soils. Soil Science Society of America Journal, 44(5), 892–898. https://doi.org/10.2136/sssaj1980.03615995004400050002x
- Van Looy, K., Bouma, J., Herbst, M., Koestel, J., Minasny, B., Mishra, U., et al. (2017). Pedotransfer functions in Earth system science: Challenges and perspectives. *Reviews of Geophysics*, 55(4), 1199–1256. https://doi.org/10.1002/2017RG000581
- Vereecken, H., Schnepf, A., Hopmans, J. W., Javaux, M., Or, D., Roose, T., et al. (2016). Modeling soil processes: Review, key challenges, and new perspectives. Vadose Zone Journal, 15(5), vzj2015.09.0131. https://doi.org/10.2136/vzj2015.09.0131
- Vereecken, H., Weynants, M., Javaux, M., Pachepsky, Y., Schaap, M., & Genuchten, M. T. v. (2010). Using pedotransfer functions to estimate the van Genuchten–Mualem soil hydraulic properties: A review. Vadose Zone Journal, 9(4), 795–820. https://doi.org/10.2136/vzj2010.0045
- Verhoef, A., & Egea, G. (2014). Modeling plant transpiration under limited soil water: Comparison of different plant and soil hydraulic parameterizations and preliminary implications for their use in land surface models. *Agricultural and Forest Meteorology*, 191, 22–32. https://doi.org/10.1016/j.agrformet.2014.02.009

Wagner, B., Tarnawski, V., Hennings, V., Müller, U., Wessolek, G., & Plagge, R. (2001). Evaluation of pedo-transfer functions for unsaturated soil hydraulic conductivity using an independent data set. *Geoderma*, *102*(3–4), 275–297. https://doi.org/10.1016/S0016-7061(01)00037-4

Webb, S. W. (2000). A simple extension of two-phase characteristic curves to include the dry region. Water Resources Research, 36(6), 1425– 1430. https://doi.org/10.1029/2000WR900057

Weihermüller, L., Lehmann, P., Herbst, M., Rahmati, M., Verhoef, A., Or, D., et al. (2021). Choice of pedotransfer functions matters when simulating soil water balance fluxes. *Journal of Advances in Modeling Earth Systems*, 13(3), e2020MS002404. https://doi.org/10.1029/2020MS002404 Weynants, M., Vereecken, H., & Javaux, M. (2009). Revisiting Vereecken pedotransfer functions: Introducing a closed-form hydraulic model.

Vadose Zone Journal, 8(1), 86–95. https://doi.org/10.2136/vzj2008.0062

 Wilson, J. P., Lam, C. S., & Deng, Y. (2007). Comparison of the performance of flow-routing algorithms used in GIS-based hydrologic analysis. *Hydrological Processes: International Journal*, 21(8), 1026–1044. https://doi.org/10.1002/hyp.6277
 Wösten, J., Lilly, A., Nemes, A., & Le Bas, C. (1999). Development and use of a database of hydraulic properties of European soils. *Geoderma*,

Wosten, J., Liny, A., Nemes, A., & Le Bas, C. (1999). Development and use of a database of hydraulic properties of European soils. *Geoderma*, 90(3–4), 169–185. https://doi.org/10.1016/S0016-7061(98)00132-3

Wösten, J., Pachepsky, Y. A., & Rawls, W. (2001). Pedotransfer functions: Bridging the gap between available basic soil data and missing soil hydraulic characteristics. *Journal of Hydrology*, 251(3–4), 123–150. https://doi.org/10.1016/S0022-1694(01)00464-4

Wu, D., Ciais, P., Viovy, N., Knapp, A. K., Wilcox, K., Bahn, M., et al. (2018). Asymmetric responses of primary productivity to altered precipitation simulated by ecosystem models across three long-term grassland sites. *Biogeosciences*, 15(11), 3421–3437. https://doi.org/10.5194/ bg-15-3421-2018

Xu, X., Medvigy, D., Powers, J. S., Becknell, J. M., & Guan, K. (2016). Diversity in plant hydraulic traits explains seasonal and inter-annual variations of vegetation dynamics in seasonally dry tropical forests. *New Phytologist*, 212(1), 80–95. https://doi.org/10.1111/nph.14009

Yang, Y., Yuan, H., & Yu, W. (2018). Uncertainties of 3D soil hydraulic parameters in streamflow simulations using a distributed hydrological model system. *Journal of Hydrology*, 567, 12–24. https://doi.org/10.1016/j.jhydrol.2018.09.042

Zacharias, S., & Wessolek, G. (2007). Excluding organic matter content from pedotransfer predictors of soil water retention. Soil Science Society of America Journal, 71(1), 43–50. https://doi.org/10.2136/sssaj2006.0098

Zhang, X., Tang, Q., Zhang, X., & Lettenmaier, D. P. (2014). Runoff sensitivity to global mean temperature change in the CMIP5 models. Geophysical Research Letters, 41(15), 5492–5498. https://doi.org/10.1002/2014GL060382

Zhang, Y., & Schaap, M. G. (2017). Weighted recalibration of the Rosetta pedotransfer model with improved estimates of hydraulic parameter distributions and summary statistics (Rosetta3). Journal of Hydrology, 547, 39–53. https://doi.org/10.1016/j.jhydrol.2017.01.004

Zhang, Y., & Schaap, M. G. (2019). Estimation of saturated hydraulic conductivity with pedotransfer functions: A review. Journal of Hydrology, 575, 1011–1030. https://doi.org/10.1016/j.jhydrol.2019.05.058

Zhang, Z., Paschalis, A., & Mijic, A. (2021). Planning London's green spaces in an integrated water management approach to enhance future resilience in urban stormwater control. *Journal of Hydrology*, 597, 126126. https://doi.org/10.1016/j.jhydrol.2021.126126

Zheng, Y., Shen, R., Wang, Y., Li, X., Liu, S., Liang, S., et al. (2020). Improved estimate of global gross primary production for reproducing its long-term variation, 1982–2017. Earth System Science Data, 12(4), 2725–2746. https://doi.org/10.5194/essd-12-2725-2020

Zhou, S., Duursma, R. A., Medlyn, B. E., Kelly, J. W., & Prentice, I. C. (2013). How should we model plant responses to drought? An analysis of stomatal and non-stomatal responses to water stress. Agricultural and Forest Meteorology, 182, 204–214. https://doi.org/10.1016/j. aerformet.2013.05.009

Zhou, S., Liang, J., Lu, X., Li, Q., Jiang, L., Zhang, Y., et al. (2018). Sources of uncertainty in modeled land carbon storage within and across three MIPs: Diagnosis with three new techniques. *Journal of Climate*, *31*(7), 2833–2851. https://doi.org/10.1175/JCLI-D-17-0357.1